Author's response to reviews

Title: Influence of gender and rising food prices on poor, pregnant women's eating and food provisioning practices in Dhaka, Bangladesh

Authors:

Adrienne V Levay (alevay@ualberta.ca)
Zubia Mumtaz (zubia.mumtaz@ualberta.ca)
Sabina Faiz Rashid (sabina@bracu.ac.bd)
Noreen Willows (noreen.willows@ualberta.ca)

Version: 2 Date: 19 September 2013

Author's response to reviews:

see attachment